Author's response to reviews

Title: Community hospitals - The place of local service provision in a modernising NHS: An integrative thematic literature review

Authors:

David Heaney (D.Heaney@abdn.ac.uk)
Corri Black (corri.black@abdn.ac.uk)
Catherine A O'Donnell (Kate.O'Donnell@clinmed.gla.ac.uk)
Cameron Stark (c.stark@abdn.ac.uk)
Edwin van Teijlingen (van.teijlingen@abdn.ac.uk)

Version: 4 Date: 20 December 2006

Author's response to reviews: see over
20 December 2006.

The BioMed Central Editorial Production Team
BMC Public Health
BioMed Central.

Authors: David Heaney, Corri Black, Catherine A O'Donnell, Cameron Stark and Edwin van Teijlingen.
Title: Community hospitals – the place of local service provision in a modernising NHS: An integrative thematic literature review.
MS: 9025837961051156

Please find attached the three files pertaining to the above document. They have been formatted according to your instructions.

If there are any problems, please do not hesitate to contact me and I will rectify them.

Thank you.

Dr Catherine O'Donnell.
Senior Lecturer in Primary Care R&D.
Tel: 0141 330 8329.
Email: Kate.O'Donnell@clinmed.gla.ac.uk